Author’s response to reviews

Title: Clinical Management of Malignant Insulinoma: A Single Institution’s Experience over Three Decades

Authors:
Jie Yu (bailubailing@163.com)
Fan Ping (pingfan6779@163.com)
Huabing Zhang (huabingzhang@sina.com)
Wei Li (WeiLi43@163.com)
Tao Yuan (yuan3750057@163.com)
Yong Fu (YongFu057@163.com)
Kai Feng (KaiFeng0695@163.com)
Weibo Xia (WeiboXia357@126.com)
Lingling Xu (llxuwsh@163.com)
Yuxiu Li (liyuxiu@medmail.com.cn)

Version: 2 Date: 30 Jul 2018

Author’s response to reviews:

1. We have the consent to publish, because all in patients in our hospital have signed a written informed consent to use their clinical data and blood samples for research and to publish related scientific findings. Furthermore, patients’ records/information were anonymized before analysis.

2. The study was approved by the Medical Ethics Committee of the PUMCH. The reference number is S-K534, and date of approval is 2017-5-18.